Author's response to reviews

Title: Democracy: The Forgotten Challenge for Bioethics in the Developing Countries

Authors:

Ghaiath M.A. Hussein (ghaiathme@gmail.com)

Version: 5 Date: 14 May 2009

Author's response to reviews: see over
May 14, 2009

The Biomed Central Editorial Team

Object: Democracy: The Forgotten Challenge for Bioethics in the Developing Countries

Ghaiath M. A. Hussein, MBBS, MHSc. (Bioethics)

Dear Editor,

Thank you for the acceptance of my manuscript for publication in your journal. I have reviewed the above manuscript according to the author’s guidelines and made the necessary changes. They are as follows:

1- I have reformulated the abstract into the following sections: Background, Discussion, and Conclusion.
2- I have re-written the title page as per the given instructions.
3- I have divided the manuscript under the same sections in the abstract (background, discussion, and conclusion)
4- I have reviewed the citation format. Now they are in square brackets before the full stop.
5- I have added the Competing Interests sections between the conclusion section and references.